Author's response to reviews

Title: Variation in incidence of breast, lung and cervical cancer and malignant melanoma of skin by socioeconomic group in England.

Authors:

Shack Lorraine (lorraine.shack@nwcis.nhs.uk)
Catrina Jordan (catrina.jordan@ic.nhs.uk)
Catherine S Thomson (catherine.thomson@wmciu.nhs.uk)
Vivian Mak (vivian.mak@kcl.ac.uk)
Henrik Moller (henrik.moller@kcl.ac.uk)

Version: 4 Date: 14 July 2008

Author's response to reviews: see over
Replies to referees’ comments

Variation in incidence of breast, lung and cervical cancer and malignant melanoma of skin by socioeconomic group in England.

Referee 3: Gwenn Menvielle

We are most grateful for these positive comments.

Page 9: Sentence is rephrased for clarity
Page 10 last paragraph: the second mention of 2006 was removed.
Page 13: Reference to published audit added and rephrasing of paragraph for clarity.

Editor request:
Sentences addressing ethical requirements have been added to the Methods section.